

Scientific Annals of Economics and Business 71 (2), 2024, 193-219

DOI: 10.47743/saeb-2024-0012





Impact of Cost of Capital on European Economic Growth: The Role of IFRS Mandatory Adoption

Ghouma Ghouma*, Hamdi Becha***, Maha Kalai***, Kamel Helali

Abstract: Since 2005, the International Financial Reporting Standards (IFRS) mandatory adoption in the European Union has played a pivotal role to reduce financing costs which has influenced positively economic growth across member states. Thus, this study examines the effect of Cost of Capital on Economic Growth under IFRS mandatory adoption in 17 European countries between 1994 and 2021 using Pooled Mean Group Autoregressive Distributed Lag (PMG-ARDL) and System Generalized Method of Moments (GMM-system) methods. The findings reveal a positive correlation between the Cost of Capital and Economic Growth under IFRS adoption. Specifically, the model estimates indicate that the Cost of Capital contributes to a 0.58% increase in Economic Growth in the PMG-ARDL framework. Moreover, the GMM-system model underscores the significance of IFRS adoption in reducing the Cost of Capital, leading to a 0.52% increase in Economic Growth. These results provide insights into the benefits of adopting international accounting standards and highlight the importance of institutional and financial factors in shaping the economic impact of adopting accounting standards.

Keywords: Cost of Capital; Economic Growth; European Countries; PMG-ARDL; GMM-system.

JEL classification: C23; M40; O11; O52.

Article history: Received 17 May 2023 | Accepted 26 May 2024 | Published online 26 June 2024

To cite this article: Ghouma, G., Becha, H., Kalai, M., Helali, K. (2024). Impact of Cost of Capital on European Economic Growth: The Role of IFRS Mandatory Adoption. *Scientific Annals of Economics and Business*, 71(2), 193-219. https://doi.org/10.47743/saeb-2024-0012.



This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License.

^{*}Faculty of Economics and Management of Sfax, University of Sfax, Tunisia; e-mail: ghoumahamouda@gmail.com.

Faculty of Economics and Management of Sfax, University of Sfax, Tunisia; e-mail: bechahamdi@yahoo.com.

Faculty of Economics and Management of Sfax, University of Sfax, Tunisia; e-mail: maha.kalai@fsegs.usf.tn.

[§] Faculty of Economics and Management of Sfax, University of Sfax, Tunisia; e-mail: kamel.helali@fsegs.usf.tn (corresponding author).

1. INTRODUCTION

Economies, investors, and lenders depend on the free flow of capital and investment between countries. International Financial Reporting Standards (IFRS) have become a global language used by investors in more than 165 countries when evaluating cross-border investments (Prather-Kinsey *et al.*, 2022). In particular, the European Union (EU) has approved a regulation that, since 2005, listed companies in the European Union, including insurance companies and banks, can prepare their consolidated financial statements by IFRS. For this reason, the International Accounting Standards Board (IASB) introduced these Standards which represent an accounting code that aims to create a single financial reporting platform worldwide (Mohsin *et al.*, 2021).

As described in Van Greuning *et al.* (2011), this regulation heralded the biggest changes in financial reporting in Europe in the last 30 years, and these changes directly affected approximately 7000 companies and indirectly affected various types of consolidated subsidiaries. IFRS is governed by the IASB, an organization whose mission is to serve the public interest by promoting long-term economic growth, confidence, and financial stability in the global economy through reliable financial information. In general, IFRS replaces Generally Accepted Accounting Principles (GAAP). For example, members of the EU and the European Economic Union (EEA) have imposed a mandatory requirement to use IFRS for listed companies in member states and non-member countries since 2005 (Mager and Meyer-Fackler, 2017; Nguyen, 2018).

According to Zeff (2005), accounting standard setting reflects the growing importance of financial accounting standards in different sectors of the economy, which has led to increasing lobbying by special interests for accounting standards with characteristics consistent with desired outcomes. Financial accounting standards affect the economy in several ways, both overall and in the distribution of income, wealth, and risk. In addition, IFRS yields significant benefits for companies and adopting countries in terms of improved transparency, reduced capital costs, improved cross-border investment, improved comparability of financial reports, and increased scrutiny by foreign analysts (IASB, 2014; De George *et al.*, 2016).

The adoption of IFRS provides quality financial information, thereby encouraging FDI (Gordon *et al.*, 2012). The information provided when using IFRS is clear, which reduces business risk. In addition, the success of countries in integrating international trade developments has made free trade an important condition for promoting their growth and development. According to Cooke and Wallace (1990), the more open an economy is to the outside world, the more it will be exposed to international pressures. These pressures reflect complex business volumes but provide a skilled workforce and workers. In addition, they represent a comparative advantage in wage costs. These advantages are likely to attract investors and thus expand their economic activities. Thus, there are significant benefits to adopting accounting standards to facilitate international transactions (Kolsi and Zehri, 2013).

In the European case, Oppong and Aga (2019) examine the impact of IFRS adoption on economic growth using the GMM method for 28 European economies over a period from 2005 to 2014. The results of the model reveal that IFRS adoption has a positive and significant impact on economic growth. Furthermore, full adoption of IFRS is significantly associated with economic growth in both developed and developing countries, while partial adoption is only significant for economic growth in developing countries. Therefore, IFRS adoption is

not the only factor that may affect the rate of economic growth, but developing countries could benefit fully from IFRS adoption if they consider that geographical, macroeconomic, and political factors influence economic growth, as these factors appear to be important in explaining economic growth in the EU.

The effects of mandatory IFRS adoption in the European Union on economic growth is a very interesting and important topic for those interested in financial reporting. The voluntary adoption of IFRS in the European Union (EU) during the 1990s was a significant step toward harmonizing accounting practices across member countries which has been mandated since 2005 and their impact on the cost of equity of firms and economic growth has been significant (Bengtsson, 2022). Specifically, IFRS has played a crucial role in lowering the cost of capital and enhancing the financial ecosystem. As a result, cross-border investment has been facilitated, and the allocation of capital optimized, contributing to overall economic growth. This underscores the critical importance of IFRS as a key driver of financial stability and prosperity within the European Union.

Given the critical importance of financial reporting to economic growth, this study provides valuable insights into the potential benefits of adopting IFRS Standards. As highlighted by Ball (2006) and Barth *et al.* (2012) have extensively explored the positive effects of IFRS on financial reporting and capital markets which has positive effects on economic growth (Ben Othman and Kossentini, 2015; Oppong and Aga, 2019; Banker *et al.*, 2021; Owusu *et al.*, 2022). However, there remains a noticeable gap in understanding the nuanced relationship between the cost of capital and economic growth specifically within the EU context post-IFRS adoption.

While prior studies acknowledge the general benefits of IFRS, a focused examination of how variations in the cost of capital, influenced by IFRS, directly contribute to or hinder economic growth in the EU is notably underexplored. Addressing this gap is crucial for policymakers, investors, and businesses as it provides insights into the mechanisms through which financial reporting standards impact the broader economic landscape, offering valuable guidance for decision-making and policy formulation in the European Union.

Thus, the contribution of this study lies in its ability to provide concrete proof of the potential benefits of the mandatory adoption of IFRS and its considerable impact on the relationship between Cost of capital and economic growth for 17 European Countries adopted these Standards between 1994 and 2021 using two econometric methods which are Pooled Mean Group Auto-Regressive Distributed Lag (PMG-ARDL) and the System Generalized Methods of Moments (GMM-system). In this context, this research aims to answer the following research question: Does the cost of capital have a favorable effect on economic growth in EU countries under the mandatory adoption of International Financial Accounting Standards (IFRS)?

Additionally, many prior investigations have predominantly relied on singular methodologies, limiting the depth of their analyses. Thus, by incorporating both methodologies concurrently, this research seeks to provide a more comprehensive and robust assessment of the relationship between the cost of capital and economic growth. Moreover, unlike previous studies that may have employed broader or less relevant timeframes, this research takes a focused approach, analyzing the period between 1994 and 2021. This allows us to directly observe and analyze the dynamic interplay between changes in accounting standards and their real-world economic consequences, specifically on the cost of capital and growth.

The remainder of this paper is organized as follows. Section 2 provides a comprehensive overview of the theoretical and empirical framework underlying the research topic by

presenting the existing literature on the relationship between mandatory IFRS adoption and economic growth, exploring the main theories, concepts, and ideas that have been advanced by researchers and experts in the field, as well as presenting the hypothesis development. Section 3 describes the sample selection, data, and empirical specifications. Section 4 presents the empirical estimations of this study. Finally, Section 5 presents some conclusions of the findings and policy implications.

2. LITERATURE REVIEW

IFRS is the abbreviation for "International Financial Reporting Standards" which are accounting standards developed by the International Accounting Standards Board (IASB). They are used as a common financial reporting language by companies and organizations around the world to ensure transparency, comparability, and consistency of financial reporting. These standards guide various aspects of financial reporting, including financial statement presentation, recognition and measurement of assets, liabilities, and equity, and disclosure requirements. IFRS also guides in specialized areas such as revenue recognition, leases, and financial instruments and is intended to ensure that financial information is accurate, reliable, and relevant to the needs of investors, creditors, and other stakeholders.

Moreover, IFRS promotes transparency and consistency in financial reporting, making it easier for investors to compare financial information between countries. This can increase cross-border investment, and therefore economic growth. In addition, by providing standardized financial information, IFRS can increase the availability of capital for companies, particularly those in emerging markets, leading to increased investment, growth, and job creation. In addition, it encourages companies to present their financial information accurately and transparently, allowing investors and stakeholders to make informed decisions about resource allocation. This can lead to more efficient use of resources, better productivity, and increased economic growth.

2.1 Theoretical framework

The macroeconomic justification for the widespread adoption of IFRS is underpinned by two key theories: the economic theory of networks by Katz and Shapiro (1985) and the neo-institutional theory elucidated by DiMaggio and Powell (1991).

In the economic theory of networks, IFRS implementation creates a global financial reporting network, promoting seamless communication and collaboration among diverse economic entities. This interconnectedness enhances market efficiency, reduces information asymmetry, and attracts international investment, contributing to overall economic growth.

Conversely, isomorphism posits that countries adopting IFRS align themselves with global standards, gaining legitimacy and signaling a commitment to transparency and best practices. This alignment attracts foreign investment and harmonizes financial reporting practices, fostering economic stability and growth.

From the economic theory of networks perspective, countries are more likely to adopt IFRS if their trade partners or countries within their geographical region have already embraced IFRS. This theory views IFRS as a product, assessing its intrinsic value and the value derived from its network. If a country shares a close economic relationship with others

that have adopted IFRS, implementing IFRS reduces domestic bias for foreign investors and facilitates multinational operations (Ramanna and Sletten, 2009).

The "autarky value of IFRS" represents its intrinsic value, considering economic and political benefits. Economic net value refers to more efficient resource allocation, while the political value represents control over the standard-setting process. IFRS standards are considered valuable for a developing country if the autarky value and the synchronization value of IFRS surpass the value of local GAAP.

For the institutional theory, they explain the more and more homogeneous organizational behavior and structure by the concept of isomorphism. According to DiMaggio and Powell (1983); DiMaggio and Powell (1991), three types of isomorphism can be used to explain the adoption of IFRS in one country. The first type is Coercive isomorphism which involves institutions forcing economic actors to align with IFRS, such as the International Monetary Fund (IMF) requiring financial reforms in exchange for aid. The second type is mimetic isomorphism encompasses the imitation of nations perceived as more legitimate and successful, potentially influenced by professional accounting organizations. The third type is the normative isomorphism which is linked to a country's education level, with higher education attainment affecting accounting practices and the shift toward IFRS (Hassan, 2008).

2.2 Empirical framework

Several studies show that the adoption of IFRS can increase investment, improve access to capital markets, improve the quality of financial reporting, and promote economic growth. Several studies examined the effect of IFRS on economic growth. Larson (1993) used a similar approach and carried out a cross-sectional study with data from 35 African countries to assess the differences in growth patterns between countries that adopted IFRS to suit their local environment, countries that adopted the standards without any adaptation, and countries that did not adopt IRFS. The study reveals that compared to complete adopters and non-adopters, countries that adopt IFRS to suit their local environment experience substantially higher economic growth.

Larson and Kenny (1995) conducted an exploratory study that empirically examined the relationships between the adoption of International Accounting Standards (IAS), the development of stock markets, and economic growth in developing countries with stock markets for 27 developing countries between 1985 and 1989, using a transnational sociological research design and partial least squares. The results indicate that there is no major association between the development of stock markets in developing countries and economic growth due to the adoption of IAS standards.

Leuz and Verrecchia (2000) conducted a study on a sample of German companies that adopted either IAS or US-GAAP Accounting Standards in their consolidated financial statements. Their results showed that companies that are committed to increasing their disclosure levels receive economic benefits. Firstly, an international communication strategy is associated with a reduction in bid-ask spreads and an increase in stock turnover, while controlling for various firm characteristics such as performance, firm size, foreign listings, and selection bias. Secondly, regarding stock price volatility, the authors demonstrated a negative association or reduction around the switch to international accounting, and there are minor differences between companies that follow US-GAAP and those that follow IAS.

Zeghal and Mhedhbi (2006) conducted a study focused on the determinants of the adoption of IAS/IFRS standards by developing countries. Their study was based on a sample of 32 developing countries that have adopted IAS standards and 32 other countries that have not. They concluded that developing countries with developed capital markets, advanced education levels, and high economic growth are more likely to adopt IFRS standards.

In another study, Akisik (2013) examined the relationship between accounting regulation, financial development, and economic growth in 51 developed and emerging market economies over the period 1997-2009 using the Generalized Method of Moments estimation techniques. This study provides evidence that accounting regulation has a significant effect on economic growth, even after controlling for several macroeconomic and socio-economic variables.

Using a sample of 50 emerging economies over a period from 2001 to 2007, Ben Othman and Kossentini (2015) found a positive association between the degree of adoption of IFRS standards and the development of the stock market. Given that it has been proven that the development of the stock market promotes economic growth in developing economies (Adjasi and Biekpe, 2006; Cooray, 2010), the adoption of IFRS standards could enhance the economic growth of a country.

In a study spanning a decade from 2005 to 2015, Özcan (2016) investigates the impact of IFRS adoption on the economic growth of 41 countries that embraced IFRS and 29 countries that did not adopt IFRS standards. Through the application of a panel data model, the regression results demonstrate a significant increase in the economic growth of countries following the adoption of IFRS and while IFRS adoption emerges as a contributing factor, other elements such as education policies, human capital, geographical factors, and political structure also influence economic development rates.

Oppong and Aga (2019) studied the influence of IFRS adoption on economic growth across 28 European economies from 2005 to 2014, using the GMM method. This modeling reveals a positive impact of IFRS adoption on economic growth. The study also shows that full adoption of IFRS is associated with economic growth in both developed and developing countries, while partial adoption is only significant for economic growth in developing countries. This study suggests that IFRS adoption may not be the only factor impacting economic growth, as other factors such as geographic, macroeconomic, and political conditions may also play a key role in explaining economic growth in the EU. Therefore, developing countries could benefit from adopting IFRS by taking into account these factors that affect economic growth.

Akisik *et al.* (2020) examine the effect of IFRS adoption on economic growth in 41 African countries over the period 1997 and 2017 and found that the IFRS adoption and economic growth is statistically insignificant but its interaction with FDI has a significant and positive effect on economic growth, suggesting that the adoption of IFRS may not be beneficial for economic growth and IFRS appears to enhance the positive impact of FDI on economic growth.

In a recent study, Banker et al. (2021) examine whether productivity improves after mandatory IFRS adoption for 141 countries between 1996 and 2013 using mandatory IFRS as a shock to the accounting regime to examine changes in country-level productivity. The authors find that countries that adopt mandatory IFRS experience significant increases in Total Factor Productivity (TFP) and labor productivity and that post-adoption productivity improvements are larger for countries without convergence to IFRS. In addition, TFP

increases more for countries that experience a greater increase in industry comparability. In addition, the new IFRS accounting regime increases economic productivity by improving information environments and facilitating internal business decisions.

More recently, Owusu *et al.* (2022) examine whether the adoption of IFRS Standards affects economic growth in developing economies, and investigate the role of institutional quality at the country level in this relationship using panel data spanning three non-overlapping years between 1996 and 2013 for 78 developing countries, and employing the two-step Generalized Method of Moments (GMM) method. Their results showed that countries that adopt IFRS experience better economic growth than those that do not. Furthermore, good institutions can moderate the link between IFRS and economic growth.

2.3 Hypothesis development

The adoption of IFRS does not directly affect the cost of equity or economic growth. However, the adoption of IFRS can indirectly impact both of these factors. IFRS can affect the cost of equity by improving the transparency and comparability of financial statements, which can reduce the perceived risk of investing in a company. As a result, investors may require a lower return on their investment, reducing the cost of equity for the company. Moreover, IFRS can indirectly impact economic growth by improving the quality of financial reporting, which can enhance the ability of investors and creditors to make informed decisions about allocating capital. This, in turn, can increase the availability of capital for investment and stimulate economic growth. Indeed, this research can be further elaborated into this fundamental hypothesis (H1) which is that the cost of capital has a positive impact on economic growth in European countries under IFRS adoption.

3. RESEARCH DESIGN

3.1 Sample and data

The widespread adoption of the International Financial Reporting Standards (IFRS) has generated intense research interest within the last two decades. As Leuz and Wysocki (2008) point out, both firm-level and macroeconomic evidence are important in evaluating the economic consequences of accounting standards or reporting regulations. Thus, we provide some evidence on the macro front by investigating the link between IFRS adoption and the economic growth of countries.

This study assesses the effect of IFRS adoption on economic growth, through the variable cost of capital. Therefore, to estimate the effect of IFRS adaptation in equity on economic growth in European countries and concerning Oppong and Aga (2019), we will consider the following model:

$$GDPG_{it} = \beta_0 + \beta_1 Activity_{it} + \beta_2 GFCF_{it} + \beta_3 INFL_{it} + \beta_4 Trade_{it} + \beta_5 EXR_{it} + \beta_6 Mean_Ke_{it} + \beta_7 IFRS_{it} + \varepsilon_{it}$$
 (1)

where

- "GDPG" is the GDP growth rate in annual percentage, this indicator is expressed in annual percentage terms and represents an important economic indicator that measures the rate at which a country's economic output is expanding or contracting over a year. It represents

the vital instrument of the economic cycle and progression in countries (Acquah and Ibrahim, 2020; Song *et al.*, 2020).

- "Activity" is the ratio of the working population to the working age population (15 years and older), the activity rate serves as a valuable tool for analyzing the level of economic activity within a given population and plays a role in understanding the labor market dynamics on a national or global scale. Human capital is regarded as a measure of the ability and skills of a labor force and it is evaluated by the formal education or the job learning accumulated experience that plays a relevant role in enabling innovation and R&D activities.
- "GFCF" (% of GDP) is the ratio of gross fixed capital formation to GDP, this indicator measures the proportion of a country's total economic output used for investment in fixed capital assets and provides insights into the level of investment relative to the economic output. Investment is the heart of economic growth and plays a vital role in its economic progression. Psycharis *et al.* (2020) and Alfredsson and Malmaeus (2019) prove that public capital investment enhances industrial productivity, boosts GDP growth, and way forward to sustainability development.
- "INFL" is the Inflation rate in percentage of the consumer price index, this indicator reflects the rate at which the general price level of goods and services is rising, leading to a decrease in the purchasing power of a currency. In a neo-classical growth model, Haslag (1997) generalized the relationship between inflation and growth in the long run by introducing money and showed that there would be negatively related if money was a complement to capital, positively related if money was a substitute to capital, and independent if money was only a medium of exchange.
- "Trade" (% of GDP) is the ratio of the sum of exports and imports divided by GDP, this indicator measures the importance of international trade to the economic output of a country and reflects the degree to which a nation is engaged in global trade activities and the impact of international commerce on its economic performance. International trade is a necessary way for a country to cope with economic globalization, and it has long been debated whether the degree of trade openness is beneficial to the economic development of all countries in the world (Kirikkaleli and Oyebanji, 2022). Trade openness has a positive effect on economic growth (Manwa et al., 2019; Ehigiamusoe and Babalola, 2021; Odebode and Oladipo, 2021; Rehman et al., 2021). Carrasco and Tovar-García (2021) investigated the trade-economic growth relationship in developing countries, using a dynamic panel data model, and found that trade-induced growth in developing countries benefits from imports of high-tech and capital goods.
- "EXR" is the exchange rate of a currency against the domestic currency, in which has profound implications for the economy, influencing various aspects of international trade, investment, and overall economic health. There is a relatively large body of literature suggesting a correlation between the real exchange rate and GDP growth. As long as productivity is higher in the traded goods sector, countries have an incentive to maintain the relative price of traded goods high enough to make it attractive to shift resources into their production (Habib *et al.*, 2017).
- "Mean_Ke" is the aggregate cost of capital by country which represents the overall cost a company or entities within an economy incur to finance their operations and investments and reflects the return required by investors for providing funds to these entities. Mean_Ke plays a crucial role in economic growth by influencing investment decisions. A lower aggregate cost of capital may encourage businesses to invest in projects and expand

operations, fostering economic growth. In a recent study, Ghouma *et al.* (2023) found that IFRS adoption reduces in cost of equity capital which supports the argument that high-quality accounting standards enhance the quality of financial reporting and positively affect firms' cost of equity capital.

- "*IFRS*" is a dichotomous variable equal to 1 when the financial statements are prepared by IFRS and 0 otherwise, the year of adoption of IFRS in Europe thus begins with 1 between 2005 and 2021 and 0 between 1994 and 2004 (Gordon *et al.*, 2012; Ben Othman and Kossentini, 2015; Oppong and Aga, 2019; Owusu *et al.*, 2022).

All the variables were collected from World Development Indicators (WDI) and DataStream and our sample is presented by 17 European countries, which are: Austria, Belgium, Denmark, Finland, France, Germany, Ireland, Italy, Luxembourg, Norway, Netherlands, Poland, Portugal, United Kingdom, Spain, Sweden, and Switzerland, between 1994 and 2021 (i.e. T=28). The parameters of the model measure the sensitivity of the variables to economic growth. We summarize all these variables in Table no. 1.

Variables **Definition** Source **GDPG** Gross Domestic Product (Annual %) WDI The ratio of the working population to the working-age population (15 WDI Activity years and older) **GFCF** Gross Fixed Capital Formation (% of GDP) WDI WDI **INFL** Inflation rate (% of Consumer Price Index) WDI Trade The sum of exports and imports (% of GDP) WDI **EXR** The exchange rate of a currency against the domestic currency The aggregate cost of capital by country DataStream Mean Ke The dichotomous variable is equal to 1 when the financial statements are **IFRS** DataStream prepared by IFRS and 0 otherwise

Table no. 1 - Variables definitions

Notes: WDI refers to World Development Indicators DataBank.

3.2 Estimation techniques

We start by presenting the methodology of the Pooled Mean Group Autoregressive Distributed Lag (PMG-ARDL) method, this technique was proposed by Pesaran *et al.* (1999) and Pesaran *et al.* (2001) to overcome the limitations of the methods of Engle and Granger (1987) and Johansen (1991).

An advantage of the PMG-ARDL model is that the short-term dynamic specifications can vary across cross sections, while the long-term coefficients must be the same. Therefore, the PMG-ARDL model is used to study the cross-sectional heterogeneous dynamic problem and evaluate the short- and long-term relationship between the variables. In addition, the statistical advantages of this method are higher efficiency, low collinearity, and higher degree of freedom parameters (Lee, 1995). Furthermore, this estimator is made under the assumption of cointegration of all variables, which must be [I(0)], [I(1)], or a mixture of [I(0)] and [I(1)].

The reason for using PMG-ARDL is that it has a relatively small sample size and is therefore less sensitive to the presence of outliers (Pesaran et al., 1999). In addition, the serial autocorrelation problem can be simultaneously corrected, and this rather lagged model has the advantage of reducing the endogeneity problem (Pesaran et al., 1999), which has been a concern in the recent literature on economic growth and its application. Therefore, we obtain

unbiased estimates of the short-term model (Harris and Sollis, 2003). The equation of the ARDL model will be the following:

$$\begin{split} \textit{GDPG}_{it} = \ \delta_0 + \delta_1 \textit{GDPG}_{it-1} + \ \delta_2 \textit{Activity}_{it-1} + \ \delta_3 \textit{GFCF}_{it-1} + \ \delta_4 \textit{INFL}_{it-1} \\ + \ \delta_5 \textit{Trade}_{it-1} + \delta_6 \textit{EXR}_{it-1} + \delta_7 \textit{Mean}_{.} \textit{Ke}_{it-1} \\ + \sum_{j=1}^p \alpha_{1j} \Delta \textit{GDPG}_{it-j} + \sum_{j=0}^q \alpha_{2j} \Delta \textit{Activity}_{it-j} \sum_{j=0}^q \alpha_{3j} \Delta \textit{GFCF}_{it-j} \\ + \sum_{j=0}^q \alpha_{4j} \Delta \textit{INFL}_{it-j} + \sum_{j=0}^q \alpha_{5j} \Delta \textit{Trade}_{it-j} + \sum_{j=0}^q \alpha_{6j} \Delta \textit{EXR}_{it-j} \\ + \sum_{i=0}^q \alpha_{7j} \Delta \textit{Mean}_{.} \textit{Ke}_{it-j} + u_{it} \end{split} \tag{2}$$

Alternatively, a dynamic error correction model (ECM) can be drawn, whose error correction term ρ_1 must be negative and significant to indicate the speed of adjustment, i.e., the time required for the variables to return to long-term equilibrium. Thus, the ECM model can be represented by the following equation:

$$\begin{split} \Delta GDPG_{it} = & \ \rho_0 + \rho_1 \hat{\varepsilon}_{it-1} + \sum\nolimits_{j=1}^p \alpha_{1j} \Delta GDPG_{it-j} + \sum\nolimits_{j=0}^q \alpha_{2j} \Delta Activity_{it-j} \\ & + \sum\nolimits_{j=0}^q \alpha_{3j} \Delta GFCF_{it-j} + \sum\nolimits_{j=0}^q \alpha_{4j} \Delta INFL_{it-j} + \sum\nolimits_{j=0}^q \alpha_{5j} \Delta Trade_{it-j} \\ & + \sum\nolimits_{j=0}^q \alpha_{6j} \Delta EXR_{it-j} + \sum\nolimits_{i=0}^q \alpha_{7j} \Delta Mean_Ke_{it-j} + \mu_{it} \end{split} \ (3)$$

where $\hat{\varepsilon}_{it-1}$ is the estimated residual error of the above Eq. 1.

To verify the results presented by ARDL, we proceed to use the System Generalized Method of Moments (GMM-system) method. This technique allows us to verify the robustness of results by checking if the estimated parameters are consistent across different models and also allows us to assess the sensitivity of their results to different assumptions. Arellano and Bover (1995) and Blundell and Bond (1998) develop a system of difference and level regressions. The instruments of difference regression are the lagged levels of the explanatory variables and the instruments of level regression are the lagged differences of the explanatory variables. These are considered appropriate instruments under the assumption that while there may be a correlation between the levels of the explanatory variables and the country-specific effect, there is no correlation between these variables in the differences and the country-specific effect.

Moreover, this technique is more efficient with an additional assumption that the first differences of instruments are uncorrelated with the fixed effects, which in turn allows the inclusion of more instruments (Roodman, 2009) and yields efficient estimates in cases where the series are close to being random walks (Blundell and Bond, 1998). The equation of the GMM-system can be written as follows:

$$\begin{split} GDPG_{it} &= \beta_0 + \beta_1 GDPG_{it-1} + \beta_2 Acticity_{it} + \beta_3 GFCF_{it} + \beta_4 INFL_{it} + \beta_5 Trade_{it} \\ &+ \beta_6 EXR_{it} + \beta_7 Mean_Ke_{it} + \beta_8 IFRS_{it} + \varepsilon_{it} \end{split} \tag{4}$$

In addition, to ensure the GMM-system estimation's consistency of Eq. 1, we employed two primary diagnostics. Firstly, we used the Arellano and Bond (1991) test to confirm that

the first differenced error term is without first- and second-order serial correlation. The null hypothesis of the p-value reported by AR(2) is that the first difference residuals have no serial correlation in the second-order. Secondly, we applied the Hansen (1982) test to identify constraints that report p-values for null assumptions and make sure that the error term should not be correlated with instruments.

4. RESULTS

4.1 Variables analysis

Before starting the analysis of the analysis of cross-section dependence, the stationarity of variables, the cointegration relationship between, and the analysis of the models, it is essential to start with the descriptive analysis taken into account for this estimation. Thus, we begin by analyzing statistically the key variables in this study.

According to Table no. 2, for the 476 observations, the variable "GDPG" is characterized by an overall mean of 2.236, a median of 2.261, and a standard deviation of 2.988. Furthermore, the minimum and maximum of this variable are equal to -10.823 and 25.176, respectively. Moreover, the distribution is completely skewed to the right (Skewness=0.299>0), and strongly platykurtic (Kurtosis=12.568> 0). Moreover, this distribution is non-normal for the whole sample and is no longer auto-correlated. Furthermore, the variable "Mean_Ke" is characterized by an overall mean of 0.014, a median of 0.008, and a standard deviation of 0.194. In addition, the minimum and maximum of this variable are equal to -3.148 and 1.317, respectively. Moreover, the distribution is completely skewed to the left (Skewness= -7.971>0), and strongly platykurtic (Kurtosis=162.014>0).

Variables **GDPG** Activity **GFCF** INFL Trade EXR Mean_Ke **IFRS** Observations 476 476 476 476 476 2.236 60.086 21.756 1.988 100.014 23.431 0.014 0.607 Mean 0.194 Standard deviation 2.988 3.439 2.517 61.861 168.806 0.489 5.862 Minimum -10.823 47.270 14.752 -4.478 40.455 0.500 -3.148 0 Maximum 25.176 74.110 53.591 32.991 388.848 1736.207 1.317 1 0.903 Median 2.261 60.175 21.599 1.812 79,441 0.008 -7.971-0.439Skewness 0.299 0.078 2.217 7.419 2.299 9.393 Kurtosis 12 568 2 826 19311 80.073 8 803 90.727 162.014 1.193 Jarque & Bera (JB) test 1823 1.085 5667 1.2e+05 1087 1.6e+05 Probability (JB) 0.000 0.581 0.0000.000 0.000 0.000Born & Breitung (BB) test 3.04 14.43 7.33 13.79 11.53 2.02 0.218 0.001 0.026 0.001 0.003 Probability BB 0.365

Table no. 2 – Descriptive statistics of variables

Notes: JB refers to Jarque and Bera (1987) normality test. BB refers to Born and Breitung (2016) serial correlation test.

4.2 Cross-Sectional Dependence

The next step, after the statistical description of variables, is the application of the dependence test as this test represents an important step in the application of panel models. The dependence test is performed to test the hypothesis of cross-sectional independence and to determine the appropriate panel unit root test to adopt. De Hoyos and Sarafidis (2006) highlight the need and importance of testing for cross-sectional dependence in dynamic panel

data. Specifically, Sarafidis and Robertson (2006) indicate that the absence of cross-sectional independence in the data can lead to inconsistency in all estimation procedures.

In more detail, cross-sectional dependence should be considered when analyzing panel data, as it can impact the validity of statistical tests and lead to inaccurate findings. Pesaran (2007) underscores the importance of accounting for this dependence, especially in unit root tests and cointegration analysis. The assumption of observational independence may be violated by cross-sectional dependence, and maintaining this assumption is crucial for reliable statistical inference. Neglecting cross-sectional dependence may result in a misleading and inconsistent interpretation of the long-term relationships between variables. Thus, to ensure the accuracy and reliability of cointegration tests, it is imperative to account for cross-sectional dependence (Banerjee *et al.*, 2004).

Therefore, the Cross-sectional Dependence (CD) test developed by Pesaran (2004) is performed. In this vein, De Hoyos and Sarafidis (2006) argue that in dynamic panels, the validity of Pesaran (2021) tests remains proven under fixed/random effect estimation, and it is therefore considered the preferred choice for cross-sectional independence analysis. To this end, we will examine different tests of dependence such as the Friedman (1937), Frees (1995, 2004), and Pesaran (2004); Pesaran *et al.* (2008) tests. These tests examine the existence of a dependent relationship between individuals. According to Table no. 3, all the tests affirm the existence of a dependency between individuals because all the probabilities of these tests are lower than 1%.

Table no. 3 – Cross-Section Dependence test results

Tests	Value	Probability	Decision
Friedman (1937)	269.511	0.000	Dependence
Frees (1995, 2004)	5.307	0.000	Dependence
Pesaran (2004)	39.964	0.000	Dependence
Pesaran et al. (2008)	37.078	0.000	Dependence

4.3 Unit root tests

In the presence of cross-sectional dependence (CD), the use of first-generation unit root tests may lead to inaccurate results. Therefore, we have opted for unit root tests that can accurately identify the stationarity properties of the variables. For this reason, we have used second-generation stationarity tests for the determination of the stationarity attribute of the variables under study. Thus, both CADF and CIPS unit root tests of Pesaran (2003); Pesaran (2007) respectively were used to determine the order of integration of these variables (Akadiri *et al.*, 2020; Onifade *et al.*, 2023).

These tests, presented in Table no. 4, have been developed to asymptotically eliminate the problem of dependence between series and have the property of being robust. Thus, to control the order of integration, second-generation panel root tests have been performed. The results of these unit root tests show that the variables GDPG, INFL, EXR, and Mean_Ke are stationary in level while the variables Activity, GFCF, and Trade and stationary in first difference. We perform also another unit root test with a break to verify these results.

GDPG INFI Variables **GFCF** Trade **EXR** Activity Mean_Ke Pesaran (2003) Panel A: In level -2.780* -3.880** -2.691 -4.086 With Constant -2.026 -2.005 -1.862 -2.787*** -2.911*** -3.822*** With Trend -2.390-2.952* -2.105 -4.125* NS NS S Decision S NS S Panel B: In first difference -3.665*** -5.331*** -4.386*** -4.838*** With Constant -3.274* -3.754* -4.319^* With Trend -4.409*** -3.407** -3.892** -4.589*** -3.881*** -5.200*** -5.486** Decision S S S \mathbf{S} S S S Pesaran (2007) Panel A: In level -1.699 With Constant -3.815 -2.094-2.077 -3.227-3.526 -4.832 -3.850*** -3.106*** -3.312*** -3.355** -2.491 -1.946 -4.736* With Trend NS Decision S NS NS S S S Panel B: In first difference -6.005** -5.945** -5.151** With Constant -4.449* -4.100° -5.356* -4.362 -6.004*** -4.613*** -4.561*** -5.602*** -4.279*** -5.057*** -6.150*** With Trend

Table no. 4 – Second-generation unit root test

Notes: *, ***, *** represent significance at 10%, 5%, 1%, respectively. NS: non-stationary. S: Stationary.

S

The results of Karavias and Tzavalis (2014) unit root test with break presented in Table no. 5 show that the series are stationary in level or first difference with the presence of some breaks in the years 1995, 1997, 1999, 2015, and 2020. As most of the variables are stationary in level or first difference, it is important to study the existence of a cointegration relation between them.

Table no. 5 – Unit root test with a break

Variables	In level	In first difference
GDPG	-28.559***(1995)	-39.166***(1995)
Activity	-4.282***(1995)	-28.710***(1995)
GFCF	-11.425***(2015)	-42.923***(2020)
INFL	-14.765***(1997)	-30.674***(2020)
Trade	-15.338***(2020)	-30.793***(2020)
EXR	-1.9e+02***(1999)	-33.512***(1995)
Mean_Ke	-37.520*** (1995)	-45.890*** (2020)

Notes: The value between brackets represents the break date. *** represents significance at 1%.

4.4 Cointegration tests

Decision

S

Since the variables are stationary in the first difference, it is necessary to find a long-term relationship (cointegration relationship) to justify the switch to the GMM approach. Therefore, we proceed to test for the existence of a long-term relationship between variables by using the panel Cointegration test. Therefore, we use the tests of Kao (1999); Pedroni (2004); Westerlund (2007) to determine the existence of a Cointegration relationship between variables.

Table no. 6 - Cointegration tests

Tests	t-Statistics	Probability	Decision
Kao (1999)	-6.227	0.000	Co-integration
Pedroni (2004)	-8.091	0.000	Co-integration
Westerlund (2007)	-2.392	0.008	Co-integration

According to the results of Table no. 6 which presents the different cointegration tests, namely that of Kao (1999); Pedroni (2004); Westerlund (2007), all the tests have a probability lower than the 5% threshold, so we can say that there is at least one cointegration relationship for all the variables of our model. Thus, the null hypothesis is strongly rejected and consequently, the variables have a long-term relationship between variables. This confirmation not only enhances the reliability of our model but also offers insights into the speed of adjustment and short-term dynamics governing the long-term relationships among the variables.

4.5 PMG-ARDL results

After testing the Cointegration relationship between variables, we can proceed to estimate the model with the PMG-ARDL method by Pesaran *et al.* (2001) suggested to test the effect of IFRS adoption on economic growth in European countries. This estimator is performed with the assumption of the existence of cointegration between all variables which must be [I(0)], [I(1)], or a mixture of [I(1)] and [I(0)].

Moreover, the serial autocorrelation problem can be corrected simultaneously and the advantage of using panel ARDL with sufficient lags is a reduction in the endogeneity problem (Pesaran *et al.*, 1999) which has been a concern in the recent literature on economic growth. The results presented in Table no. 7 show that the most appropriate model is represented by an ARDL(1,0,0,0,0,0,0) with 1 lag and with a trend that has a maximum F-statistic (F-test = 12.971 with a significance level of 0.0003, higher than Pesaran *et al.* (2001) critical value of 3.79 at 5%).

The results of the short-term model reveal that the error correction term, ECT_{it-1} , is both statistically significant and negative. This proves the existence of a cointegrating relationship between the variables in the short-term model. Specifically, the estimated ECT_{it-1} value is equal to -0.476, indicating that the rate at which the long-term equilibrium adjusts in response to the imbalances caused by the short-term shocks of the previous period is 47.6%. Briefly, this coefficient represents the strength of recall, meaning that 47.6% of the imbalance between real and desired economic growth is adjusted in response to economic shocks. Therefore, a shock to economic growth is fully absorbed after 2 years, 1 month, and 6 days.

In the short term, we observe that the cost of capital has a positive impact on the economic growth of EU countries. Therefore, the adoption of IFRS has an immediate effect on the important role of the cost of capital in increasing economic growth in European countries. The IFRS standards as soon as they are adopted give a positive sign in the short term for the European economies. In addition, the adoption of IFRS, in the short term, improves the level of transparency of financial information and reduces the information asymmetry in countries which leads to an improvement in FDI flows (Gordon *et al.*, 2012; Lungu *et al.*, 2017; Turki *et al.*, 2017a; Turki *et al.*, 2017b; Abad *et al.*, 2018; Yousefinejad *et al.*, 2018).

Table no. 7 - Pooled Mean Group ARDL estimation

Variables	Coefficient	Standard-deviation	t-statistic	Probability
Short-term estimation of ARDL $(1,0,0,0,0,0,0)$: dependent variable: Δ GDPG _{it}				
GDPG _{it-1}	-1.255	0.114	-11.013	0.000
Activity _{it-1}	0.423	0.205	2.065	0.034
GFCF _{it-1}	0.193	0.065	2.950	0.003
Tradeit-1	0.203	0.061	3.316	0.001
EXR _{it-1}	0.457	1.380	0.331	0.740
INFL _{it-1}	-0.837	0.217	-3.856	0.000
Mean_Keit-1	0.733	0.351	2.090	0.037
$\Delta GDPG_{it-1}$	0.053	0.080	0.663	0.508
Δ Activity _{it}	0.569	0.351	1.625	0.104
$\Delta GFCF_{it}$	0.611	0.197	3.102	0.002
$\Delta Trade_{it}$	0.407	0.050	8.144	0.000
ΔEXR_{it}	0.825	1.902	0.434	0.665
$\Delta INFL_{it}$	-0.462	0.179	-2.580	0.010
ΔMean Keit	2.812	5.089	0.552	0.581
Trend	-0.247	0.112	-2.215	0.027
ECT _{it-1}	-0.476	0.081	-5.894	0.000
Constant	15.320	24.694	0.062	0.535
Long-term estimation of ARDL(1,0,0,0,0,0,0): dependent variable: GDPG _{it}				
Constant	12.204	19.724	0.618	0.526
Activityit	0.337	0.162	2.071	0.038
GFCF _{it}	0.153	0.051	2.986	0.002
Tradeit	0.161	0.045	3.515	0.000
EXR_{it}	0.364	1.102	0.330	0.741
$INFL_{it}$	-0.667	0.184	-3.621	0.000
Mean_Keit	0.584	0.279	2.089	0.037

According to the long-term estimation of the PMG-ARDL model presented above, the results show that every 1% increase in the variable Activity increases growth by 0.34%. This factor is also considered to be a very important factor for an economy and it helps to provide a highly skilled and innovative workforce that can use limited resources efficiently, thus increasing per capita income. Effective human capital also attracts FDI, which stimulates economic growth (Intisar et al., 2020). Moreover, any increase in gross fixed capital formation increases economic growth by 0.15%. The rate of capital formation can play a main role in promoting economic growth. In fact, for an open economy, the link between investment and economic growth is very important. In other words, a higher ratio of gross fixed capital formation has a positive impact on growth so growth opportunities also lead to further increases in domestic investment (De Long and Summers, 1991, 1992). Indeed, an increase in public investment increases output and therefore has long-term growth effects thus productive government activity and long-term economic growth allow for a better resulting welfare allocation. Furthermore, according to Barro (1990), government spending as a public good in the production function of individual firms increases the rate of return to private capital and therefore stimulates growth (Irmen and Kuehnel, 2009). Similarly, along with public investment spending, adequate private capital inflows accelerate economic growth in economies. Thus, private capital investment helps to stimulate economic growth (Choong et al., 2010).

For the Trade variable, every 1% increase has a positive effect on economic growth of 0.16% and this result reveals that more liberalized trade regimes are accompanied by higher export and GDP growth rates. Trade openness is also considered likely to improve technology because a large international market can provide technological spillovers, economies of scale in research and development, and higher profits for innovators (Katz and Shapiro, 1985; Romer, 1990; Grossman and Helpman, 1991; Rivera-Batiz and Romer, 1991). The link between trade openness and growth can exist through investment and technology. For the first link, trade openness favors investment because the trade sector is more capital-intensive than the non-trade sector. For the second link, the production of investment goods uses imported intermediate goods to some extent, and competition in the international market for machinery and capital goods lowers the price of capital (Baldwin and Seghezza, 1996).

Concerning the inflation variable, any 1% increase in inflation reduces economic growth by 0.67%. Indeed, inflation is a harmful factor that affects output growth because it discourages exports, and savings and public spending become inefficient and increases taxes, which reduces purchasing power, increases uncertainty, discourages investment, and thus reduces productivity and economic growth. In other words, high inflation can lead to uncertainty about the future benefits of investment projects and thus reduce aggregate output (Azam and Khan, 2022).

The exchange rate exhibits a positive but statistically non-significant impact on economic growth, with a 1% increase in the exchange rate correlating to a 0.36% rise in economic growth. This positive association suggests that while exchange rate fluctuations contribute to economic expansion, other intricate factors concurrently wield significant influence. Effective policy measures, resilient fiscal strategies, and a stable global economic environment may be collaboratively acting to alleviate potential adverse effects of currency movements.

Moreover, it also seems that companies adapt to changes in exchange rates quite well, employing tactics that capitalize on the advantages while minimizing the drawbacks. This positive but not significant link highlights how complex the economic environment is, with various elements interacting to shape growth, and gives a picture of the situation that goes beyond significance.

According to the results, the cost of capital on economic growth has a positive and significant influence that a 1% increase in the cost of capital leads to a 0.58% increase in economic growth. Therefore, the adoption of IFRS in the EU has a positive influence on economic growth because it has created a higher level of confidence for investors and the transparency of financial reporting promotes increased foreign investment in host country companies. As a result, foreign investors are concerned about the costs and risks associated with investing in companies, and the adoption of IFRS helps to alleviate these concerns by creating greater confidence in financial reporting.

The adoption of IFRS has proven to be an effective strategy to improve the reliability of financial reporting, reduce information asymmetry among users of accounting information, and reduce the costs and risks associated with foreign direct investment (FDI) flows. By reducing the costs and risks of doing business in a foreign country, foreign investors gain greater confidence and assurance in investing outside their home country. In addition, improved corporate governance and reduced information asymmetry further enhance investor confidence.

Therefore, it can be concluded that IFRS adoption improves FDI, as evidenced by studies such as Gordon *et al.* (2012); Jinadu *et al.* (2016); Lungu *et al.* (2017). These studies suggest that IFRS adoption creates greater transparency and comparability in financial reporting, which leads to greater foreign investor confidence and encourages more foreign investment in host country firms. Foreign Direct Investment (FDI) plays a crucial role in stimulating economic growth and an increase in FDI flows leads to a significant improvement in economic growth (Borensztein *et al.*, 1998; Gudaro *et al.*, 2012; Pegkas, 2015; Bermejo Carbonell and Werner, 2018; Sokang, 2018; Song *et al.*, 2020). Given the economic growth benefits associated with increased FDI inflows, many countries, particularly developing nations, are actively seeking ways to increase FDI inflows. These efforts reflect a growing recognition of the important role that FDI can play in economic development and improved living standards.

Table no. 8 - Diagnostic tests

Tests	Value (p-value)	Decision
ARCH test	1.096 (0.295)	No heteroscedasticity
Serial autocorrelation LM-test	0.227 (0.634)	No autocorrelation
Ramsey test	0.320 (0.572)	No specification errors

The diagnostic tests conducted on the ARDL model, as detailed in Table no. 8, reveal a robust validation of the model outcomes. The results affirm that the error terms within the short-term model exhibit freedom from heteroscedasticity, indicating consistent variability. Moreover, the absence of serial correlation and specification errors confirms the reliability of the model results and underscores the validity of the ARDL model and confidence in its capacity to provide accurate results.

4.6 GMM-system results

To verify the results obtained above, we need to apply the GMM-system method of Arellano and Bover (1995) and Blundell Blundell and Bond (1998), to check the long-term impact of variables on economic growth.

According to the GMM model results presented in Table no. 9, the results show that every 1% increase in Activity variable increases economic growth by 0.24%. Human capital is also considered a fundamental source of economic growth because it improves the level of total productivity and potential earnings of the labor force. Investment in human capital improves the quality of education and leads to self-sustaining growth. Moreover, investment in technological research results in new knowledge and technological development, and therefore investment in human capital is a key factor for sustainable economic growth (Lucas, 1988; Romer, 1990). In addition, human capital is also valued by skills, qualifications the ability to create new products, and experience of the workforce through specialization and division of labor, improving basic education, vocational training, encouraging self-employment, and creating business opportunities (Intisar et al., 2020).

Moreover, any increase in gross fixed capital formation increases economic growth by 0.36 percent. This is because the investments will increase production capacity and have a positive impact on employment. These new jobs will generate wage income, which in turn will increase the stock of distributed income, which is used to provide goods and services, and thus to buy and serve. Therefore, firms make profits because they own more which

increases output in the domestic market and national income (Goldsmith, 2008; Leduc and Wilson, 2014; Scandizzo and Pierleoni, 2020).

Variables	Coefficient	Standard-deviation	t-statistic	Probability
GDPG _{it-1}	-0.106	0.034	-2.94	0.010
Activityit	0.238	0.057	4.16	0.001
GFCF _{it}	0.358	0.090	3.99	0.001
$INFL_{it}$	0.563	0.078	7.24	0.000
Tradeit	0.024	0.004	5.75	0.000
EXR_{it}	-0.001	0.0002	-3.05	0.008
Mean_Keit	0.520	0.227	2.29	0.022
IFRS _{it}	1.907	0.312	6.11	0.000
Constant	4.891	2.925	1.67	0.114

Table no. 9 – Two-step GMM-system estimation

Furthermore, investment is central to economic growth and plays a critical role in its economic progression. In other words, appropriate public financing and resource allocation policy accelerates economic activities. Moreover, public capital investment improves industrial productivity, stimulates GDP growth, and paves the way for sustainable development (Alfredsson and Malmaeus, 2019; Psycharis *et al.*, 2020).

For the inflation variable INFL, the results show that an increase of 1% in the inflation rate increases economic growth by 0.56%. Indeed, economic growth can benefit greatly from moderate inflation. In other words, modest inflation can stimulate economic growth by lowering real interest rates and promoting investment and spending. This idea supports the Keynesian viewpoint by highlighting the role that aggregate demand plays in promoting economic expansion. Additionally, moderate inflation can reduce debt loads, which in turn makes it simpler for governments and consumers to service loans and reallocate resources to more beneficial purposes.

Moreover, a 1% increase in the Trade variable exerts a positive influence, leading to a 0.024% improvement in economic growth. Indeed, trade openness and economic growth have a positive relationship with economic growth and there is a bidirectional causality between them. Our results are consistent with those (Yanikkaya, 2003; Wang *et al.*, 2004; Borrmann *et al.*, 2006; Armstrong *et al.*, 2008; Chang *et al.*, 2009; Oppong and Aga, 2019; Akisik *et al.*, 2020; Owusu *et al.*, 2022). Liberalization can provide useful insight into the gains from trade liberalization because it increases manufacturing output growth and can affect growth (Lee, 1995).

In addition, trade liberalization facilitates knowledge and technology transfer, which in many endogenous growth models improves productivity and growth (Grossman and Helpman, 1991) by increasing the size of the market and thus the demand for goods and services in the economy as a whole, which indirectly forces producers to generate a high level of output to meet unexpected demand. Concerning the relationship between the degree of trade openness and economic growth, the conventional view is that trade openness has growth-enhancing effects. Therefore, countries with outward-looking economies tend to grow faster than inward-looking economies. The explanation is that openness to trade generally expands a country's trade opportunities, increases the rate of technology transfer and diffusion, and improves the efficiency of resource allocation in an economy, which in turn improves productivity and growth (Wang et al., 2004; Borrmann et al., 2006; Abad et al., 2018).

Additionally, the exchange rate has a low negative and significant on the economic growth of European countries. The changes in exchange rates have a negative effect on economic growth because they disrupt trade and investment and a volatile or depreciating currency can reduce a country's export competitiveness by raising the cost of its goods and services for foreign consumers which decreases exports has an impact on the country's overall economic output. Moreover, a volatile currency rate creates uncertainty, which discourages foreign direct investment and hinders enterprises' long-term planning.

Furthermore, capital inflows are hampered by investors' reluctance to participate in crossborder initiatives, which reduces the amount of money available for profitable projects and reduces the possibility of long-term economic growth. Thus, the adverse consequences of fluctuating exchange rates surpass their direct influence on commerce, infiltrating many aspects of the economy and generating an atmosphere less favorable for strong and consistent expansion.

Taking into account the impact of the cost of capital on economic growth, any 1% increase in the cost of capital increases economic growth by 0.52%. Our results are consistent with those (Choong *et al.*, 2010; Chen *et al.*, 2015; Francis *et al.*, 2016; Brown and Martinsson, 2019; Amel-Zadeh *et al.*, 2020; Golshan *et al.*, 2023).

It is revealed that the adoption of IFRS leads to a positive and significant increase in the GDP growth rate. This result is consistent with the institutional theory that IFRS adoption has a significant positive effect on economic growth, which is similar to Zehri and Abdelbaki (2013); Oppong and Aga (2019); Akisik *et al.* (2020), who found that IFRS adoption stimulates economic growth. The higher the level of harmonization with IFRS, the greater the development of adopting countries. The results also support the old institutional theory hypothesis that IFRS (in this case, mandatory adoption) is an institutional infrastructure that could stimulate growth and development (North, 1990).

The consistency of the estimated growth model parameters is based on the results of diagnostic tests of the estimated GMM-system dynamic model, presented in Table no. 10, which indicate the presence of first-order AR(1) autocorrelation and the absence of second-order AR(2) autocorrelation in the model residuals. In addition, the Hansen (1982) test is used to check the validity of the instruments to ensure that the model is not mis-specified.

Table no. 10 - Arellano and Bond autocorrelation and over-identification tests

Tests	Value	Probability
AR(1)	-2.91	0.004
AR(2)	-1.26	0.104
Hansen	15.910	0.460

Our results show that there is no serial correlation in the disturbances in the second-order AR(2), while the Hansen (1982) test demonstrates that the instruments used in the GMM specification are valid. This indicates that the estimated parameters are reliable and can be used to draw valid conclusions about the relationship between the variables in the growth regression model.

In other words, the absence of serial correlation within the perturbations of the secondorder AR(2) model provides evidence of the temporal independence of the errors, which is crucial for ensuring the integrity of our model. These findings are consistent with the results of the Hansen (1982) test, which not only confirms the absence of serial correlation but also attests to the validity of the instruments incorporated in the System Generalized Method of Moments (GMM-system) specification which is crucial as to resolve endogeneity problems in the growth regression model.

5. CONCLUSION AND POLICY IMPLICATIONS

This study examines the impact of the cost of capital on economic growth under the mandatory adoption of IFRS in 17 European countries between 1994 and 2021 using GMM-system and PMG-ARDL approaches. The results show that the cost of capital has a positive effect on economic growth under IFRS adoption. In other words, the model estimates show that the cost of capital increases economic growth by 0.58% in the PMG-ARDL. In the GMM-system model, the results show that IFRS adoption has important to lower the cost of capital, which increases the economic growth by 0.52%.

Before the adoption of IFRS, investors often faced greater uncertainty due to the absence of regulated reporting methods. In the 1990s period, the adoption of IFRS by corporations was voluntary, and as a result, financial reporting transparency improved significantly, boosting investor confidence and ultimately reducing the cost of capital. Since 2005, mandatory adoption has reinforced these positive trends by extending the benefits of lower capital costs to a wider range of enterprises. By facilitating access to capital for businesses, promoting strategic investments, and creating a more dynamic and effective business climate, these combined efforts have been crucial in boosting economic growth.

The positive effect of mandatory IFRS Standards on economic growth can be attributed to several factors. Firstly, IFRS improves the comparability and transparency of financial statements, enabling investors to make more informed decisions. Secondly, IFRS provides a more accurate picture of a company's financial situation, which helps to reduce the risk of financial anomalies and fraud. Finally, the adoption of IFRS encourages foreign investment by creating a level playing field for international investors.

IFRS reduces the knowledge gap between investors and companies by strengthening accountability and by providing both capital producers and capital holders with important information that holds both parties accountable. The implementation of these standards reduces investment risk, creating more investment opportunities for investors worldwide. Furthermore, IFRS provides high-quality accounting information that enables effective financial decision-making.

The adoption of IFRS can promote the harmonization of financial information across different countries and support the development of national accounting standards based on IFRS. This can reduce the cost and complexity of financial statement preparation for companies operating in multiple countries. Moreover, IFRS can improve financial stability by providing investors with better information about the financial situation of companies, which increases the comparability and reliability of financial statements, improves credit rating quality, and reduces the risk of financial crisis.

It should be noted that the majority of variables have positive and significant effects on economic growth, except inflation in the PMG-ARDL approach and the exchange rate in the GMM approach (negative sign).

Overall, our results using both approaches (PMG-ARDL & GMM) validate hypothesis H1 above, which assumes that the cost of capital has a positive impact on economic growth in European countries under IFRS adoption.

To foster further economic growth, the adoption of IFRS allows policymakers to invest in training programs that help companies better understand and apply IFRS. This can help address the complexity and interpretation issues associated with IFRS. In addition, policymakers can seek to harmonize accounting practices across European countries for more consistent implementation of IFRS and strengthen enforcement mechanisms to ensure that companies comply with IFRS by including more severe penalties for non-compliance and better monitoring and control of financial reporting. In addition, policymakers can help small businesses implement IFRS by including access to education and training programs as well as financial support to cover implementation costs.

As with any research, several research limitations should be acknowledged. Firstly, the complexity of economic systems and the multitude of factors influencing growth make it challenging to isolate and attribute causality solely to the cost of capital. Additionally, the heterogeneity in economic structures and regulatory environments across European nations may introduce variations in the observed effects, and the availability and quality of data, especially about the cost of capital calculations, could pose limitations on the accuracy and comparability of findings. Moreover, the dynamic nature of IFRS adoption and potential changes in regulatory frameworks over time may influence the long-term effects, requiring a nuanced interpretation of the results.

ORCID

Maha Kalai https://orcid.org/0000-0002-6234-7427 Hamdi Becha https://orcid.org/0000-0002-7480-7769 Kamel Helali https://orcid.org/0000-0001-7159-9557

References

- Abad, D., Cutillas-Gomariz, M. F., Sánchez-Ballesta, J. P., & Yagüe, J. (2018). Does IFRS mandatory adoption affect information asymmetry in the stock market? *Australian Accounting Review*, 28(1), 61-78. http://dx.doi.org/10.1111/auar.12165
- Acquah, A. M., & Ibrahim, M. (2020). Foreign direct investment, economic growth and financial sector development in Africa. *Journal of Sustainable Finance & Investment*, 10(4), 315-334. http://dx.doi.org/10.1080/20430795.2019.1683504
- Adjasi, C. K., & Biekpe, N. B. (2006). Stock market development and economic growth: The case of selected African countries. *African Development Review*, 18(1), 144-161. http://dx.doi.org/10.1111/j.1467-8268.2006.00136.x
- Akadiri, A. C., Gungor, H., Akadiri, S. S., & Bamidele-Sadiq, M. (2020). Is the causal relation between foreign direct investment, trade, and economic growth complement or substitute? The case of African countries. *Journal of Public Affairs*, 20(2), e2023. http://dx.doi.org/10.1002/pa.2023
- Akisik, O. (2013). Accounting regulation, financial development, and economic growth. *Emerging Markets Finance & Trade*, 49(1), 33-67. http://dx.doi.org/10.2753/REE1540-496X490103
- Akisik, O., Gal, G., & Mangaliso, M. P. (2020). IFRS, FDI, economic growth and human development: The experience of Anglophone and Francophone African countries. *Emerging Markets Review*, 45, 100725. http://dx.doi.org/10.1016/j.ememar.2020.100725
- Alfredsson, E. C., & Malmaeus, J. M. (2019). Real capital investments and sustainability-The case of Sweden. Ecological Economics, 161, 216-224. http://dx.doi.org/10.1016/j.ecolecon.2019.04.008

- Amel-Zadeh, A., Faasse, J., Li, K., & Meeks, G. (2020). Stewardship and Value Relevance in Accounting for the Depletion of Purchased Goodwillgoodwill. *Routledge Studies in Accounting*. http://dx.doi.org/10.2139/ssrn.2306584
- Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The Review of Economic Studies*, 58(2), 277-297. http://dx.doi.org/10.2307/2297968
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. *Journal of Econometrics*, 68(1), 29-51. http://dx.doi.org/10.1016/0304-4076(94)01642-D
- Armstrong, S. P., Drysdale, P., & Kalirajan, K. (2008). Asian trade structures and trade potential: an initial analysis of South and East Asian trade. SSRN. http://dx.doi.org/10.2139/ssrn.1767686
- Azam, M., & Khan, S. (2022). Threshold effects in the relationship between inflation and economic growth: Further empirical evidence from the developed and developing world. *International Journal of Finance & Economics*, 27(4), 4224-4243. http://dx.doi.org/10.1002/ijfe.2368
- Baldwin, R., & Seghezza, E. (1996). Trade-induced investment-led growth. *NBERG Working Paper*. http://dx.doi.org/10.3386/w5582
- Ball, R. (2006). International Financial Reporting Standards (IFRS): Pros and cons for investors. *Accounting and Business Research*, 36(1), 5-27. http://dx.doi.org/10.1080/00014788.2006.9730040
- Banerjee, A., Marcellino, M., & Osbat, C. (2004). Some cautions on the use of panel methods for integrated series of macroeconomic data. *The Econometrics Journal*, 7(2), 322-340. http://dx.doi.org/10.1111/j.1368-423X.2004.00133.x
- Banker, R., Huang, R., Li, Y., & Zhao, S. (2021). Do accounting standards matter for productivity? *Production and Operations Management*, 30(1), 68-84. http://dx.doi.org/10.1111/poms.13257
- Barro, R. J. (1990). Government spending in a simple model of endogeneous growth. *Journal of Political Economy*, 98(5), 103-125. http://dx.doi.org/10.1086/261726
- Barth, M. E., Landsman, W. R., Lang, M., & Williams, C. (2012). Are IFRS-based and US GAAP-based accounting amounts comparable? *Journal of Accounting and Economics*, 54(1), 68-93. http://dx.doi.org/10.1016/j.jacceco.2012.03.001
- Ben Othman, H., & Kossentini, A. (2015). IFRS adoption strategies and theories of economic development: Effects on the development of emerging stock markets. *Journal of Accounting in Emerging Economies*, 5(1), 70-121. http://dx.doi.org/10.1108/JAEE-02-2012-0006
- Bengtsson, M. (2022). National Adoptions of IFRS. *Accounting Perspectives* http://dx.doi.org/10.5772/intechopen.103679
- Bermejo Carbonell, J., & Werner, R. A. (2018). Does foreign direct investment generate economic growth? A new empirical approach applied to Spain. *Economic Geography*, 94(4), 425-456. http://dx.doi.org/10.1080/00130095.2017.1393312
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1), 115-143. http://dx.doi.org/10.1016/S0304-4076(98)00009-8
- Borensztein, E., De Gregorio, J., & Lee, J. W. (1998). How does foreign direct investment affect economic growth? *Journal of International Economics*, 45(1), 115-135. http://dx.doi.org/10.1016/S0022-1996(97)00033-0
- Born, B., & Breitung, J. (2016). Testing for serial correlation in fixed-effects panel data models. *Econometric Reviews*, 35(7), 1290-1316. http://dx.doi.org/10.1080/07474938.2014.976524
- Borrmann, A., Busse, M., & Neuhaus, S. (2006). Institutional quality and the gains from trade. *Kyklos*, 59(3), 345-368. http://dx.doi.org/10.1111/j.1467-6435.2006.00336.x
- Brown, J. R., & Martinsson, G. (2019). Does transparency stifle or facilitate innovation? *Management Science*, 65(4), 1600-1623. http://dx.doi.org/10.1287/mnsc.2017.3002

- Carrasco, C. A., & Tovar-García, E. D. (2021). Trade and growth in developing countries: The role of export composition, import composition and export diversification. *Economic Change and Restructuring*, 54, 919-941. http://dx.doi.org/10.1007/s10644-020-09291-8
- Chang, R., Kaltani, L., & Loayza, N. V. (2009). Openness can be good for growth: The role of policy complementarities. *Journal of Development Economics*, 90(1), 33-49. http://dx.doi.org/10.1016/j.jdeveco.2008.06.011
- Chen, L., Ng, J., & Tsang, A. (2015). The effect of mandatory IFRS adoption on international cross-listings. *The Accounting Review*, 90(4), 1395-1435. http://dx.doi.org/10.2308/accr-50982
- Choong, C. K., Yusop, Z., & Law, S. H. (2010). Private capital flows to developing countries: The role of the domestic financial sector. *Journal of the Asia Pacific Economy*, 15(4), 509-529. http://dx.doi.org/10.1080/13547860.2010.516173
- Cooke, T. E., & Wallace, R. O. (1990). Financial disclosure regulation and its environment: A review and further analysis. *Journal of Accounting and Public Policy*, 9(2), 79-110. http://dx.doi.org/10.1016/0278-4254(90)90013-P
- Cooray, A. (2010). Do stock markets lead to economic growth? *Journal of Policy Modeling*, 32(4), 448-460. http://dx.doi.org/10.1016/j.jpolmod.2010.05.001
- De George, E. T., Li, X., & Shivakumar, L. (2016). A review of the IFRS adoption literature. *Review of Accounting Studies*, 21, 898-1004. http://dx.doi.org/10.1007/s11142-016-9363-1
- De Hoyos, R. E., & Sarafidis, V. (2006). Testing for cross-sectional dependence in panel-data models. *The Stata Journal*, 6(4), 482-496. http://dx.doi.org/10.1177/1536867X0600600403
- De Long, J. B., & Summers, L. H. (1991). Equipment investment and economic growth. *The Quarterly Journal of Economics*, 106(2), 445-502. http://dx.doi.org/10.2307/2937944
- De Long, J. B., & Summers, L. H. (1992). Macroeconomic policy and long-run growth. Policies for Long-Run Economic Growth, 77, 93-128.
- DiMaggio, P., & Powell, W. W. (1991). The New Institutionalism in Organizational Analysis: University of Chicago Press.
- DiMaggio, P. J., & Powell, W. W. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*, 48(2), 147-160. http://dx.doi.org/10.2307/2095101
- Ehigiamusoe, K. U., & Babalola, S. J. (2021). Electricity consumption, trade openness and economic growth in developing countries: A disaggregated approach. *The Singapore Economic Review*, 1-28. http://dx.doi.org/10.1142/s0217590821500715
- Engle, R. F., & Granger, C. W. (1987). Co-integration and error correction: Representation, estimation, and testing. *Econometrica*, 55(2), 251-276. http://dx.doi.org/10.2307/1913236
- Francis, J. R., Huang, S. X., & Khurana, I. K. (2016). The role of similar accounting standards in cross-border mergers and acquisitions. *Contemporary Accounting Research*, 33(3), 1298-1330. http://dx.doi.org/10.1111/1911-3846.12176
- Frees, E. W. (1995). Assessing cross-sectional correlation in panel data. *Journal of Econometrics*, 69(2), 393-414. http://dx.doi.org/10.1016/0304-4076(94)01658-M
- Frees, E. W. (2004). Longitudinal and Panel Data: Analysis and Applications in the Social Sciences: Cambridge University Press. http://dx.doi.org/10.1017/CBO9780511790928
- Friedman, M. (1937). The use of ranks to avoid the assumption of normality implicit in the analysis of variance. *Journal of the American Statistical Association*, 32(200), 675-701. http://dx.doi.org/10.1080/01621459.1937.10503522
- Ghouma, G., Becha, H., Kalai, M., Helali, K., & Ertz, M. (2023). Do IFRS Disclosure Requirements Reduce the Cost of Equity Capital? Evidence from European Firms. *Journal of Risk and Financial Management*, 16(8), 374. http://dx.doi.org/10.3390/jrfm16080374
- Goldsmith, P. D. (2008). Economics of soybean production, marketing, and utilization *Soybeans* (pp. 117-150): AOCS Press. http://dx.doi.org/10.1016/B978-1-893997-64-6.50008-1
- Golshan, N., Khurana, I. K., & Silva, F. B. G. (2023). Financial transparency, labor productivity, and real wages. Labor Productivity, and Real Wages. http://dx.doi.org/10.2139/ssrn.4028993

- Gordon, L. A., Loeb, M. P., & Zhu, W. (2012). The impact of IFRS adoption on foreign direct investment. *Journal of Accounting and Public Policy*, 31(4), 374-398. http://dx.doi.org/10.1016/j.jaccpubpol.2012.06.001
- Grossman, G. M., & Helpman, E. (1991). Trade, knowledge spillovers, and growth. *European Economic Review*, 35(2-3), 517-526. http://dx.doi.org/10.1016/0014-2921(91)90153-A
- Gudaro, A. M., Chhapra, I. U., & Sheikh, S. A. (2012). Impact of Foreign Direct Investment on Economic Growth: A Case Study of Pakistan. *Journal of Management and Social Sciences*, 8(2), 22-30.
- Habib, M. M., Mileva, E., & Stracca, L. (2017). The real exchange rate and economic growth: Revisiting the case using external instruments. *Journal of International Money and Finance*, 73, 386-398. http://dx.doi.org/10.1016/j.jimonfin.2017.02.014
- Hansen, L. P. (1982). Large sample properties of generalized method of moments estimators. *Econometrica*, 50(4), 1029-1054. http://dx.doi.org/10.2307/1912775
- Harris, R., & Sollis, R. (2003). Applied time series modelling and forecasting: Wiley.
- Haslag, J. H. (1997). Output, growth, welfare, and inflation: a survey. Economic Review-Federal Reserve Bank of Dallas, 11-21.
- Hassan, M. K. (2008). The development of accounting regulations in Egypt: Legitimating the International Accounting Standards. *Managerial Auditing Journal*, 23(5), 467-484. http://dx.doi.org/10.1108/02686900810875299
- IASB. (2014). Project summary July 2014.
- Intisar, A. R., Yaseen, M. R., Kousar, R., Usman, M., & Makhdum, M. S. A. (2020). Impact of trade openness and human capital on economic growth: A comparative investigation of Asian countries. *Sustainability (Basel)*, *12*(7), 2930. http://dx.doi.org/10.3390/su12072930
- Irmen, A., & Kuehnel, J. (2009). Productive government expenditure and economic growth. *Journal of Economic Surveys*, 23(4), 692-733. http://dx.doi.org/10.1111/j.1467-6419.2009.00576.x
- Jarque, C. M., & Bera, A. K. (1987). A test for normality of observations and regression residuals. International Statistical Review/Revue Internationale de Statistique, 55(2), 163-172.
- Jinadu, O., Ojeka, S. A., & Ogundana, O. M. (2016). IFRS adoption and foreign direct investment: Evidence from Nigerian quoted firms. *Mediterranean Journal of Social Sciences*, 7(2), 99-105. http://dx.doi.org/10.5901/mjss.2016.v7n2p99
- Johansen, S. (1991). Estimation and hypothesis testing of cointegration vectors in Gaussian vector autoregressive models. *Econometrica*, 59(6), 1551-1580. http://dx.doi.org/10.2307/2938278
- Kao, C. (1999). Spurious regression and residual-based tests for cointegration in panel data. *Journal of Econometrics*, 90(1), 1-44. http://dx.doi.org/10.1016/S0304-4076(98)00023-2
- Karavias, Y., & Tzavalis, E. (2014). Testing for unit roots in short panels allowing for a structural break. Computational Statistics & Data Analysis, 76, 391-407. http://dx.doi.org/10.1016/j.csda.2012.10.014
- Katz, M. L., & Shapiro, C. (1985). Network externalities, competition, and compatibility. The American Economic Review, 75(3), 424-440.
- Kirikkaleli, D., & Oyebanji, M. O. (2022). Consumption-based carbon emissions, trade, and globalization: An empirical study of Bolivia. *Environmental Science and Pollution Research International*, 29(20), 29927-29937. http://dx.doi.org/10.1007/s11356-022-18495-6
- Kolsi, M., & Zehri, F. (2013). The determinants of IAS/IFRS adoption by emergent countries. *In Working paper, Emirates College of Technology*.
- Larson, R. K. (1993). International accounting standards and economic growth: an empirical investigation of their relationship in Africa. Research in third world Accounting, 2(24), 165-179.
- Larson, R. K., & Kenny, S. Y. (1995). An empirical analysis of international accounting standards, equity markets, and economic growth in developing countries. *Journal of International Financial Management* & *Accounting*, 6(2), 130-157. http://dx.doi.org/10.1111/j.1467-646X.1995.tb00054.x

- Leduc, S., & Wilson, D. (2014). Infrastructure Spending as Fiscal Stimulus: Assessing the Evidence. *Revista de Economia Institucional*, *5*(1), 1-24.
- Lee, J. W. (1995). Capital goods imports and long-run growth. *Journal of Development Economics*, 48(1), 91-110. http://dx.doi.org/10.1016/0304-3878(95)00015-1
- Leuz, C., & Verrecchia, R. E. (2000). The economic consequences of increased disclosure. *Journal of Accounting Research*, 38, 91-124. http://dx.doi.org/10.2307/2672910
- Leuz, C., & Wysocki, P. D. (2008). Economic consequences of financial reporting and disclosure regulation: A review and suggestions for future research. *SSRN*. http://dx.doi.org/10.2139/ssrn.1105398
- Lucas, R. E. (1988). On the mechanics of economic development. *Journal of Monetary Economics*, 22(1), 3-42. http://dx.doi.org/10.1016/0304-3932(88)90168-7
- Lungu, C. I., Caraiani, C., & Dascălu, C. (2017). The impact of IFRS adoption on foreign direct investments: Insights for emerging countries. Accounting in Europe, 14(3), 331-357. http://dx.doi.org/10.1080/17449480.2017.1374546
- Mager, F., & Meyer-Fackler, M. (2017). Mergers and acquisitions in Germany: 1981–2010. *Global Finance Journal*, 34, 32-42. http://dx.doi.org/10.1016/j.gfj.2017.03.004
- Manwa, F., Wijeweera, A., & Kortt, M. A. (2019). Trade and growth in SACU countries: A panel data analysis. *Economic Analysis and Policy*, 63, 107-118. http://dx.doi.org/10.1016/j.eap.2019.05.003
- Mohsin, M., Nurunnabi, M., Zhang, J., Sun, H., Iqbal, N., Iram, R., & Abbas, Q. (2021). The evaluation of efficiency and value addition of IFRS endorsement towards earnings timeliness disclosure. International Journal of Finance & Economics, 26(2), 1793-1807. http://dx.doi.org/10.1002/ijfe.1878
- Nguyen, T. L. A. (2018). Diversification and bank efficiency in six ASEAN countries. *Global Finance Journal*, 37, 57-78. http://dx.doi.org/10.1016/j.gfj.2018.04.004
- North, D. C. (1990). A transaction cost theory of politics. *Journal of Theoretical Politics*, 2(4), 355-367. http://dx.doi.org/10.1177/0951692890002004001
- Odebode, A., & Oladipo, O. S. (2021). Does Meaningful Relationship Exist Between Trade Liberalization and Economic Growth? A Case Study of a Small Open Economy. *Advances in Management and Applied Economics*, 11(2), 63-79. http://dx.doi.org/10.47260/amae/1125
- Onifade, S. T., Haouas, I., & Alola, A. A. (2023). Do natural resources and economic components exhibit differential quantile environmental effects? *Natural Resources Forum*, 47(3). http://dx.doi.org/10.1111/1477-8947.12289
- Oppong, C., & Aga, M. (2019). Economic growth in European Union: Does IFRS mandatory adoption matter? *International Journal of Emerging Markets*, 14(5), 792-808. http://dx.doi.org/10.1108/IJOEM-01-2018-0010
- Owusu, G. M. Y., Suppiah, S. D. K., Saat, N. A. M., & Law, S. H. (2022). IFRS adoption and economic growth in developing economies. *International Journal of Economics and Accounting*, 11(1), 73-98. http://dx.doi.org/10.1504/IJEA.2022.119613
- Özcan, A. (2016). Assessing the effects of IFRS adoption on economic growth: A cross country study. Artvin Çoruh Üniversitesi Uluslararası Sosyal Bilimler Dergisi, 2(2), 65-80.
- Pedroni, P. (2004). Panel cointegration: Asymptotic and finite sample properties of pooled time series tests with an application to the PPP hypothesis. *Econometric Theory*, 20(3), 597-625. http://dx.doi.org/10.1017/S0266466604203073
- Pegkas, P. (2015). The impact of FDI on economic growth in Eurozone countries. *Journal of Economic Asymmetries*, 12(2), 124-132. http://dx.doi.org/10.1016/j.jeca.2015.05.001
- Pesaran, M. H. (2003). Estimation and Inference in Large Heterogenous Panels with Cross Section Dependenc. *CESifo Working Paper*(869).
- Pesaran, M. H. (2004). General diagnostic tests for cross section dependence in panels. *University of Cambridge, Faculty of Economics, Cambridge Working Papers in Economics, 0435*.
- Pesaran, M. H. (2007). A simple panel unit root test in the presence of cross-section dependence. *Journal of Applied Econometrics*, 22(2), 265-312. http://dx.doi.org/10.1002/jae.951

- Pesaran, M. H. (2021). General diagnostic tests for cross-sectional dependence in panels. *Empirical Economics*, 60, 13-50. http://dx.doi.org/10.1007/s00181-020-01875-7
- Pesaran, M. H., Shin, Y., & Smith, R. J. (2001). Bounds testing approaches to the analysis of level relationships. *Journal of Applied Econometrics*, 16(3), 289-326. http://dx.doi.org/10.1002/jae.616
- Pesaran, M. H., Shin, Y., & Smith, R. P. (1999). Pooled mean group estimation of dynamic heterogeneous panels. *Journal of the American Statistical Association*, 94(446), 621-634. http://dx.doi.org/10.1080/01621459.1999.10474156
- Pesaran, M. H., Ullah, A., & Yamagata, T. (2008). A bias-adjusted LM test of error cross-section independence. *The Econometrics Journal*, 11(1), 105-127. http://dx.doi.org/10.1111/j.1368-423X.2007.00227.x
- Prather-Kinsey, J., De Luca, F., & Phan, H. T. P. (2022). Improving the global comparability of IFRS-based financial reporting through global enforcement: A proposed organizational dynamic. *International Journal of Disclosure and Governance*, 19(3), 330-351. http://dx.doi.org/10.1057/s41310-022-00145-5
- Psycharis, Y., Tselios, V., & Pantazis, P. (2020). The contribution of cohesion funds and nationally funded public investment to regional growth: Evidence from Greece. *Regional Studies*, 54(1), 95-105. http://dx.doi.org/10.1080/00343404.2018.1525696
- Ramanna, K., & Sletten, E. (2009). Why do countries adopt International Financial Reporting Standards? *Harvard Business School Accounting & Management Unit Working Paper*.
- Rehman, A., Radulescu, M., Ma, H., Dagar, V., Hussain, I., & Khan, M. K. (2021). The impact of globalization, energy use, and trade on ecological footprint in pakistan: Does environmental sustainability exist? *Energies*, 14(17), 5234. http://dx.doi.org/10.3390/en14175234
- Rivera-Batiz, L. A., & Romer, P. M. (1991). International trade with endogenous technological change. *European Economic Review*, 35(4), 971-1001. http://dx.doi.org/10.1016/0014-2921(91)90048-N
- Romer, P. M. (1990). Endogenous technological change. *Journal of Political Economy*, 98(5), 71-102. http://dx.doi.org/10.1086/261725
- Roodman, D. (2009). How to do xtabond2: An introduction to difference and system GMM in Stata. *The Stata Journal*, 9(1), 86-136. http://dx.doi.org/10.1177/1536867X0900900106
- Sarafidis, V., & Robertson, D. (2006). On the impact of cross section dependence in short dynamic panel estimation: University of Cambridge.
- Scandizzo, P. L., & Pierleoni, M. R. (2020). Short and long-run effects of public investment: Theoretical premises and empirical evidence. *Theoretical Economics Letters*, 10(4), 834-867. http://dx.doi.org/10.4236/tel.2020.104050
- Sokang, K. (2018). The impact of foreign direct investment on the economic growth in Cambodia: Empirical evidence. *International Journal of Innovation and Economic Development*, 4(5), 31-38. http://dx.doi.org/10.18775/ijied.1849-7551-7020.2015.45.2003
- Song, Y., Chen, B., Tao, R., Su, C. W., & Peculea, A. D. (2020). Does bilateral political relations affect foreign direct investment? Economic research. *Ekonomska Istrazivanja*, 33(1), 1485-1509. http://dx.doi.org/10.1080/1331677X.2020.1755880
- Turki, H., Wali, S., & Boujelbene, Y. (2017a). IFRS mandatory adoption effect on the information asymmetry: Immediate or delayed? *Australasian Accounting. Business and Finance Journal*, 11(1), 55-77. http://dx.doi.org/10.14453/aabfj.v11i1.5
- Turki, H., Wali, S., & Boujelbène, Y. (2017b). Adoption obligatoire des IFRS et asymétrie d'information dans le contexte français: Effet modérateur de l'endettement. *La Revue Gestion et Organisation*, 9(1), 63-78. http://dx.doi.org/10.1016/j.rgo.2017.01.001
- Van Greuning, H. V., Scott, D., & Terblanche, S. (2011). International Financial Reporting Standards. World Bank Publications. Retrieved from https://documents1.worldbank.org/curated/fr/520891468139494304/pdf/International-financial-reporting-standards-a-practical-guide-sixth-edition.pdf
- Wang, C., Liu, X., & Wei, Y. (2004). Impact of openness on growth in different country groups. *World Economy*, 27(4), 567-585. http://dx.doi.org/10.1111/j.0378-5920.2004.00614.x

- Westerlund, J. (2007). Testing for error correction in panel data. Oxford Bulletin of Economics and Statistics, 69(6), 709-748. http://dx.doi.org/10.1111/j.1468-0084.2007.00477.x
- Yanikkaya, H. (2003). Trade openness and economic growth: A cross-country empirical investigation. *Journal of Development Economics*, 72(1), 57-89. http://dx.doi.org/10.1016/S0304-3878(03)00068-3
- Yousefinejad, M., Ahmad, A., Salleh, F., Rahim, R. A., & Azam, H. (2018). The mediating effect of information asymmetry on IFRS and foreign direct investment. *International Journal of Economics and Management*, 12(2), 641-656.
- Zeff, S. A. (2005). The Evolution of U.S. GAAP: The Political Forces Behind Professional Standards. *The CPA Journal*, 75(1), 19-27.
- Zeghal, D., & Mhedhbi, K. (2006). An Analysis of the Factors Affecting the Adoption of International Accounting Standards by Developing Countries. *The International Journal of Accounting*, 41, 373-386. http://dx.doi.org/10.1016/j.intacc.2006.09.009
- Zehri, C., & Abdelbaki, A. (2013). Does adoption of international accounting standards promote economic growth in developing countries. *International Open Journal of Economics*, 1(1), 1-13.